Author's response to reviews

Title: Early rectal stenosis following stapled rectal mucosectomy for hemorrhoids

Authors:

Dr Sven Petersen (petersen-sv@khdf.de)
Gunter Hellmich (hellmich-gu@khdf.de)
Dietrich Schumann (dietrich_schumann@web.de)
Anja Schuster (schuster-an@khdf.de)
Klaus Ludwig (ludwig-kl@khdf.de)

Version: 5 Date: 8 May 2004

Dear Iratxe,
I tried harder to format the manuscript as requested in your previous email. The tables are now added to the text.
I'm somehow surprised about your question whether to publish the article in BMC Medicine or BMC Surgery. From my understanding it should be placed in BMC Surgery since the problems, which are discussed in the paper, are surgical details.
With best regards
Sven Petersen